Author's response to reviews

Title: Secular changes in height, weight and body mass index in Hong Kong Children

Authors:

Hung-kwan HK So (sohk@cuhk.edu.hk)
Edmund AS EAS Nelson (tony-nelson@cuhk.edu.hk)
Albert M AM Li (albertmli@cuhk.edu.hk)
Eric E Wong (mcwong@cuhk.edu.hk)
Joseph J Lau (jlau@cuhk.edu.hk)
Georgia S GS Guldan (gsguldan@cuhk.edu.hk)
Kwok-hang KH Mak (drkhmak@dh.gov.hk)
Youfa Y Wang (ywang@jhsph.edu)
Tai-fai TF Fok (taifaifok@cuhk.edu.hk)
Rita Yn Tz RYT Sung (yntzsung@cuhk.edu.hk)

Version: 3 Date: 29 August 2008

Author's response to reviews:

Revisions requested have been done i.e.

Abstract- Background section has been elaborated to place the study in the context of the current knowledge in the field.

Abbreviations have been placed after the Conclusions.

Competing interests - Have written that the authors declare that they have no competing interests.